Author's response to reviews

Title: Detecting imipenem resistance in Acinetobacter baumannii by automated systems (BD Phoenix, Microscan WalkAway, Vitek 2); high error rates with Microscan WalkAway

Authors:

Canan Kulah (canankulah@yahoo.com)
Elif Aktas (drelifaktas@yahoo.com)
Fusun Comert (fusbeg@yahoo.com)
Nagihan Ozlu (drozlu2003@yahoo.com)
Isin Akyar (iakyar@acibademlabmed.com.tr)
Handan Ankarali (hankarali@yahoo.com)

Version: 6 Date: 6 March 2009

Author's response to reviews:

Dear Editor,

Final manuscript pending formatting checks are done. Small typographical changes are made.

Best Regards,

Dr Canan Kulah